Author’s response to reviews

Title: An Australian Aboriginal birth cohort: a unique resource for a life course study of an indigenous population. A study protocol

Authors:

Dr Susan Sayers (sue@menzies.edu.au)
Dorothy Mackerras (dorothy@menzies.edu.au)
Gurmeet Singh (gurmeet@menzies.edu.au)
Ingrid Bucens (ingbucens@hotmail.com)
Mrs Kathyn Flynn (pharlap@octa4.net.au)
Alison Reid (alisonr@dph.uwa.edu.au)

Version: 4 Date: 25 Feb 2003

The tables have been changed to follow the figure legends